Author's response to reviews

Title: SSRI’s and other antidepressant use during pregnancy and potential neonatal adverse effects: Impact of a public health advisory and subsequent reports in the news media.

Authors:

Adrienne Einarson Ms (einarson@sickkids.ca)
Anne-Marie Schachtschneider Ms (a.k.schachtschneider@students.uu.nl)
Roland Halil Mr (roland.halil@utoronto.ca)
Enka Bollano Dr (enkabollano@hotmail.com)
Gideon Koren Dr (gkoren@sickkids.ca)

Version: 4 Date: 20 May 2005

Author's response to reviews:

letter already sent